Author's response to reviews

Title: Endoscopic application of n-Butyl-2 Cyanoacrylate (Histoacryl) on oesophagojejunal anastomotic leak: a case report.

Authors:

Manousos- Georgios Pramateftakis (mpramateftakis@hotmail.com)
Georgios Vrakas (georgiosvrakas@gmail.com)
Ioannis Kanellos (ik@hol.gr)
Ioannis Mantzoros (mantzor-@otenet.gr)
Stamatis Angelopoulos (con@the.forthnet.gr)
Efthymios Eleftheriades (gvr156@hotmail.com)
Charalampos Lazarides (charlaz@med.auth.gr)

Version: 4 Date: 14 November 2010

Author's response to reviews:

Dear Sir/Madam,

Please find below the changes applied to our document:

1. The list of authors in the manuscript is exactly as they are in the submission system (forename, surname).
2. The main title ends with ': a case report'.
3. All unnecessary capitals in the main title have been removed.
4. All colored highlighting and the running title have been removed.
5. The bullets points from the abstract have been removed as well.
6. The keywords have been removed as requested.
7. The section Acknowledgements has been removed as well.